Author's response to reviews

Title: Use of intravitreal bevacuzimab in a patient with a Von Hippel Lindau associated retinal haemangioblastoma of the optic nerve head

Authors:

Timothy A de Klerk (timothydeklerk@yahoo.com)
David HW Steel (David.Steel@chs.northy.nhs.uk)

Version: 2 Date: 2 June 2007

Author's response to reviews:

Title, Abstract, conclusion, competing interests, case overview added as per Journal of Medical Case Reports Recommendations.